Author’s response to reviews

Title: Does Accreditation Stimulate Change? A Study of the Context and the Impact of the Accreditation Process on Canadian Healthcare Organizations

Authors:

Marie-Pascale Pomey (marie-pascale.pomey@umontreal.ca)
Louise Lemieux-Charles (l.lemieux.charles@utoronto.ca)
François Champagne (francois.champagne@umontreal.ca)
Doug Angus (Angus@management.uottawa.ca)
Abdo Shabah (ashabah@santepub-mtl.qc.ca)
André-Pierre Contandriopoulos (andre-pierre.contandriopoulos@umontreal.ca)

Version: 5 Date: 18 March 2010

Author’s response to reviews:

Good morning,

We have reviewed the copy-edits to the manuscript and Figure 1 and all are acceptable to us. We have also responded to the copy-editor’s two comments in the manuscript. We note, however, that we were not sent a copy-edited version of Table 1. We have therefore uploaded clean versions of the copy-edited manuscript and Figure 1, but await instruction as regards Table 1.